\documentclass[a4paper,twocolumn,twoside,10pt]{article}
\usepackage{graphicx,amsmath,amssymb,amsthm}

\makeatletter \oddsidemargin-.25in \evensidemargin-.25in
\makeatother \topmargin-0.75in \textwidth7in \textheight9.5in

% Place your Latex macros here -----------------------------
% THEOREMS -------------------------------------------------------
\newtheorem{thm}{Theorem}[section]
\newtheorem{cor}{Corollary}[section]
\newtheorem{prob}{Problem}[section]
\newtheorem{lem}{Lemma}[section]
\newtheorem{prop}{Proposition}[section]
\theoremstyle{definition}
\newtheorem{defn}{Definition}[section]
\newtheorem{rem}{Remark}[section]
\newtheorem{ex}{Example}[section]

% MATH -----------------------------------------------------------
\newcommand{\norm}[1]{\left\Vert#1\right\Vert}
\newcommand{\trnorm}[1]{\left\Vert#1\right\Vert_{\mathbf{tr}}}
\newcommand{\abs}[1]{\left\vert#1\right\vert}
\newcommand{\set}[1]{\left\{#1\right\}}
\newcommand{\Real}{\mathbb R}
\newcommand{\eps}{\varepsilon}
\newcommand{\To}{\longrightarrow}
\newcommand{\BX}{\mathbf{B}(X)}
\newcommand{\tr}[1]{\text{tr}\left[#1\right]}
\newcommand{\maxeig}[1]{\mathbf{\lambda_{\max}}\left(#1\right)}
\newcommand{\mineig}[1]{\mathbf{\lambda_{\min}}\left(#1\right)}
\newcommand{\e}[1]{\text{E}\left[#1\right]}
\newcommand{\diag}[1]{\text{diag}\left\{#1\right\}}
\newcommand{\A}{\mathcal{A}}


% Running Head----------------------------------------------------------------
\markboth{$~$ \hfill {\rm Author(s) Name} \hfill $~$} {$~$ \hfill
{\rm Short Title of the Paper} \hfill$~$}

\pagestyle{myheadings} 
%\setcounter{page}{151}


\begin{document}
\begin{thebibliography}{1}
\bibitem{Arnold}L.~Arnold, \emph{Stochastic Differential Equations: Theory
and Applications}, John Wiley and Sons, 1974.
\bibitem{Boyd}S.~Boyd, L.~E.~Ghaoui, E.~Feron, and V.~Balakrishnan,
\emph{Linear Matrix Inequalities in System and Control Theory},
SIAM, Philadephia, 1994.
\bibitem{Cao}Y.~Y.~Cao, Y.~X.~Sun, and C.~Cheng, ``Delay-dependent robust stabilization of uncertain systems with multiple state
delays,'' \emph{IEEE Transactions on Automat. Control}, vol. 43, pp.
1608-1612, 1998.
\bibitem{Chen}W.-H.~Chen, Z.~H.~Guan, X.~Lu, ``Delay-dependent exponential stability of uncertain stochastic
systems with multiple delays: an LMI approach,'' \emph{Systems
Control Lett.}, vol. 54, pp. 547-555, 2005.
\bibitem{Cheng}
D.~Cheng, ``Stabilization of planar switched systems'',
\emph{Systems Control Lett.} vol. 51, pp. 79-88, 2004.
\bibitem{daaf}
J.~Daafouz, P.~Riedinger, C.~Iung, ``Stability analysis and control
synthesis for switched systems: a switched Lyapunov function
approach'', \emph{IEEE Trans. Automat. Control}, vol. 47, pp.
1883-1887, 2002.
\bibitem{Fridman}E.~Fridman, ``New Lyapunov-Krasovskii functionals for stability of linear retarded and neutral type
systems,'' \emph{Systems Control Lett.}, vol. 43, pp. 309-319, 2001.
\bibitem{Fridman2}E.~Fridman, ``An improved stabilization method for linear time-delay
systems,'' \emph{IEEE Transactions on Automatic Control}, vol. 47,
pp. 1931-1937, 2002.
\bibitem{Fridman3}E.~Fridman, ``Delay-dependent stability and $H_\infty$ control: Constant and time-varying
delays,'' \emph{Int. J. Control}, vol. 76, pp. 48-60, 2003.
\bibitem{Gahinet}P.~Gahinet, A.~Nemirovski, A.~Laub, and M.~Chilali,
\emph{LMI Control Toolbox User's Guide}, Natick, MA: The Mathworks,
Inc., 1995.
\bibitem{Gu}K.~Gu, ``An integral inequality in the stability problem
of time-delay systems,'' \emph{Proc. IEEE Conf. Decision Control},
Sydney, Australia, pp. 2805-2810, Dec. 2000.
\bibitem{Hale}J.~Hale and S.~M.~V.~Lunel, \emph{Introduction to Functional Differential
Equations}, Springer-Verlag, New York, 1993.
\bibitem{Han}Q.~L.~Han, K.~Gu, and X.~Yu, ``An improved estimate of robust stability bound
of time-delay systems with norm-bounded uncertainty,'' \emph{IEEE
Trans. Automat. Control}, vol. 48, pp. 1629-1634, 2003.
\bibitem{Hasminskii}R.~Z.~Has'minskii, \emph{Stochastic Stability of
Differential Equations}, Sijthoff and Noordhoff (traslation of the
Russian edition, Moscow, Nauka 1969), 1980.
\bibitem{He}Y.~He, M.~Wu, J.~H.~She, and G.~P.~Liu, ``Delay-dependent
robust stability criteria for uncertain neutral systems with mixed
delays,'' \emph{Sys. Control Lett.}, vol. 51, pp. 57-65, 2004.
\bibitem{Hespanha}J.~P.~Hespanha and A.~S.~Morse, ``Stability of switched systems with average dwell
time,'' \emph{Proceedings of the 38th IEEE Conference on Decision
and Control}, vol. 3, pp. 2655-2660, 1999.
\bibitem{Hes}J.~Hespanha, D.~Liberzon, D.~Angeli, and E.~D.~Sontag,
``Nonlinear norm-observability notions and stability of switched
systems,'' \emph{IEEE Trans. Automat. Control}, vol. 50, pp. 154-168,
2005.
\bibitem{Hu}B. Hu, X. Xu, P. J. Antsaklis, and A. N. Michel,
``Robust stabilizing control laws for a class of second-order switched
systems,'' \emph{Systems Control Lett.}, vol. 38, pp. 197-207, 1999.
\bibitem{Jun}J.~Liu, X.~Liu, and W.-C.~Xie, ``Delay-dependent robust control for uncertain switched systems with time-delay,'' \emph{Nonlinear
Analysis: Hybrid Systems}, vol. 2, pp. 81-95, 2008.
\bibitem{Kwon}O.~M.~Kwon and J.~H.~Park, ``On improved delay-dependent
robust control for uncertain time-delay systems,'' \emph{IEEE Trans.
Automat. Control}, vol. 49, pp. 1991-1995, 2004.
\bibitem{Kolmanovskii}V.~B.~Kolmanovskii, V.~B.~Nosov, \emph{Stability of
Functional Differential Equations}, Academicn Press, 1986.
\bibitem{Kolmanovskii2}V.~B.~Kolmanovskii and J.-P.~Richard, ``Stability of some
linear systems with delays,'' \emph{IEEE Trans. Automat. Control},
vol. 44, pp. 984-989, 1999.
\bibitem{Kushner}H.~J.~Kushner, \emph{Stochastic Stability and Control},
Academic Press, 1967.
\bibitem{Li0}X.~Li and C.~E.~De~Souza, ``Criteria for robust stability and
stabilization of uncertain linear system with state delay,''
\emph{Automatica}, vol. 33, pp. 1657-1662, 1997.
\bibitem{Li1}X.~Li and C.~E.~De~Souza, ``Delay-dependent robust stability and stabilization of
uncertain linear delay systems: A linear matrix approach,''
\emph{IEEE Trans. Automat. Control}, vol. 40, pp. 1144-1148, 1997.
\bibitem{Li}X.~Li and C.~E.~De~Souza, ``Delay-dependent robust $H_\infty$ control of uncertain linear state-delayed
systems,'' \emph{Automatica}, vol. 35, pp. 1313-1321, 1999.
\bibitem{Lizg}Z.~G.~Li, C.~Y.~Wen, and Y.~C.~Soh, ``Stabilization of a class of switched systems
via designing switching laws,'' \emph{IEEE Trans. Automat. Control}
vol. 46, pp. 665-670, 2001.
\bibitem{Liberzon}D.~Liberzon,
\emph{Switching in Systems and Control}, Boston, MA: Birkhauser,
2003.
\bibitem{Man}J.~L.~Mancilla-Aguilar and R.~A~Garc\'ia,
``A converse Lyapunov theorem for nonlinear switched systems,''
\emph{Systems Control Lett.}, vol. 41, pp. 67-71, 2000.
\bibitem{Matveev}A.~Matveev and A.~Savkin, \emph{Qualitative Theory of Hybrid Dynamical
Systems}, Birkh\"auser Boston, 2000.
\bibitem{Mao}X.~Mao, ``Robustness of exponential stability of
stochastic differential delay equations,'' \emph{IEEE Trans.
Automat. Control}, vol. 41, pp. 442-447, 1996.
\bibitem{Maobook}X.~Mao, \emph{Stochastic Differential Equations and their
Applications}, Chichester: Horwood Pub., 1997.
\bibitem{Mao2}X.~Mao, N.~Koroleva, and A.~Rodkina, ``Robust stability of uncertain stochastic differential delay
equations,'' \emph{Systems Control Lett.}, vol. 35, pp. 325-336,
1998.
\bibitem{Mill}
G.~Millerioux and J.~Daafouz, ``Input independent chaos synchronization
of switched systems,'' \emph{IEEE Trans. Automat. Control}, vol. 49,
pp. 1182-1187, 2004.
\bibitem{Moon}Y.~S.~Moon, P.~Park, W. H. Kwon, and Y. S. Lee,
``Delay-dependent robust stabilization of uncertain state-delayed
systems,'' \emph{Int. J. Control}, vol. 74, pp. 1447-1455, 2001.
\bibitem{Nic}S.-L.~Niculescu, ``On delay dependent stability under model trasformaion of some neutral linear
systems,'' \emph{Int. J. Control}, vol. 74, pp. 609-617, 2001.
\bibitem{Orlov}Y. Orlov, ``Extended invariance principle for nonautonomous switched
systems,'' \emph{IEEE Trans. Automat. Control}, vol. 48, pp.
1448-1452, 2003.
\bibitem{Orlov2}Y. Orlov, ``Finite time stability and robust control synthesis of uncertain
switched systems,'' \emph{SIAM J. Control Optim.}, vol. 43, pp.
1253-1271, 2004.
\bibitem{Park1}P.~Park, ``A delay-dependent stability criterion for
systems with uncertain time-invariant delays,'' \emph{IEEE
Transactions on Automatic Control}, vol. 44, pp. 876-877, 1999.
\bibitem{Park2}P.~Park, Y.~S.~Moon, and W.~H.~Kwon, ``A delay-dependent
robust stability criterion for uncertain time-delay systems,'' In
\emph{Proceedings of the American Control Conference}, Philadephia,
USA, pp. 1963--1964, 1998.
\bibitem{Schaft}A.~van der Schaft and H.~Schumacher, \emph{An Introduction to Hybrid Dynamical Systems},
Lecture Notes in Control and Information Sciences 251,
Springer-Verlag, 2000.
\bibitem{Verriest}E.~I.~Verriest and P.~Florchinger, ``Stability of stochastic
systems with uncertain time delays,'' \emph{Systems Control Lett.},
vol. 24, pp. 41--47, 1995.
\bibitem{Liu}H.~Xu, X.~Liu, and K.~L.~Teo, ``Robust $H_\infty$ stabilisation
with definite attenuance of an uncertain impulsive switched
system,'' \emph{ANZIAM J.}, vol. 46, pp. 471-484, 2005.
\bibitem{Xu0}S.~Xu and T.~Chen, ``Robust $H\sb \infty$ control for uncertain stochastic systems with
state delay,'' \emph{IEEE Trans. Automat. Control}, vol. 47, pp.
2089--2094, 2002.
\bibitem{Xu}S.~Xu, P.~Shi, Y.~Chu, and Y.~Zou,
``Robust stochastic stabilization and $H_\infty$ control of
uncertain neutral stochastic time-delay systems,'' \emph{J. Math.
Anal. Appl.}, vol. 314, pp. 1-16, 2006.
\bibitem{Yue}D.~Yue and S.~Won, ``Delay-dependent robust stability of stochastic systems with time delay and nonlinear
uncertainties,'' \emph{Elect. Lett.}, vol. 37, pp. 992-993, 2001.
\end{thebibliography}

\def \thepage {}
\date{}

% Title----------------------------------------------------------------
\title{\begin{flushleft}
\noindent {\small {\it Proceedings of EPNACS 2011 within ECCS'11
\\Emergent Properties in Natural and Artificial Complex Systems
\\Vienna, Austria - September 15, 2011\\[5mm]
% Copyright $\copyright$ 2009 Watam Press \hfill http://www.watam.org/JNSA/\\[5.0mm]
 }}
% \noindent {\small {\it Journal of Nonlinear Systems and Applications
% (2009)
% 151--154\\
%  Copyright $\copyright$ 2009 Watam Press \hfill http://www.watam.org/JNSA/\\[5.0mm]}}
\end{flushleft}\Large\bf \uppercase{Robust exponential stability of uncertain switched stochastic systems with
time-delay} }
%\author{\Large\bf Author Index}
\author{Author1, Author2 and Author3
 \thanks{Author1 and Author2 are with  Department of
Applied Mathematics, University of Waterloo, Canada. E-mails:
jnsa@uwaterloo.ca, john@uwaterloo.ca}
\thanks{ Author3 is with  Department of Civil and Environmental Engineering,
University of Waterloo, Canada.  E-mail: david@uwaterloo.ca}
\thanks{Manuscript received April 19, 2009; revised January 11, 2010.}}
 \maketitle

% Abstract----------------------------------------------------------------


{\footnotesize \noindent {\bf Abstract.}  The abstract should be
informative, precise and not exceed
200 words. Please do not include equations, tables and references in the abstract. \\
{\bf Keywords.} At least 5 key words should be provided.}

\vskip.2in

% Contents----------------------------------------------------------------

% ----------------------------------------------------------------
\section{Introduction}
Authors are required to state clearly the contributions of the paper
in the Introduction. There should be some survey of relevant
literature. Manuscripts should be submitted through the web:\\

http://www.watam.org/JNSA/ \\

Submission of a manuscript to this journal by the authors implies
that the paper has not been previously published in any language and
in any form, has not been copyrighted or submitted simultaneously
for publication elsewhere, and that the copyright for the article
will be transferred to the publisher upon acceptance of the article.
The submitted electronic files will not be returned, so the authors
are advised to retain a copy of the original article and related
material. The corresponding authors' e-mail addresses must be
included in the manuscript.

Please make sure that all equations and figures do not run off the
margins. Figures and Tables should be placed as part of the text, or
called by standard latex commands, with descriptive captions and
should be numbered consecutively.

The following is taken from a paper. ``...''


 Stochastic dynamic modeling plays an
essential role in numerous physics and engineering problems. It can
be applied wherever random properties of a dynamical system have to
be considered. Most of the emphasis is placed on the stability
analysis of the stochastic dynamical systems (see
\cite{Arnold,Hasminskii,Kushner}). Moreover, in many applications,
the physical or chemical processes are governed by more than one
dynamics: the dynamics change among a family of choices with respect
to time $t$ or state $x$. Such processes are often described by
switched systems and have been studied extensively in recent years
(see \cite{Cheng, daaf, Hes, Hu, Lizg, Liberzon,Liu,Man, Matveev,
Mill, Orlov, Orlov2, Schaft}). Time-delay and uncertainties are two
main causes for instability of dynamical systems (see
\cite{Hale,Kolmanovskii}). Numerous studies have been carried out on
stability analysis and stabilization of time-delay systems and
uncertain systems (see
\cite{Cao,Chen,Fridman,Fridman2,Fridman3,Gu,Han,He,Kwon,Li0,Li,Li1,
Mao,Mao2,Moon,Park1,Park2,Verriest,Xu0,Xu,Yue}), some of which have
been done in the scope of stochastic systems or switched systems. To
the best knowledge of the authors, few work has been done for
switched stochastic systems with both uncertainties and time-delay.
In addition, the average dwell time scheme associated with the
multiple Lyapunov functions has been proved to be an important
method for switched systems in \cite{Hespanha} (see also
\cite{Liberzon}). However, these pioneering works have only been
conducted for switched systems without delays and stochastic
perturbations, although switched systems with delays or stochastic
perturbations are in fact very common in practice. Moreover, as long
as the stability of a time-delay system is considered, there exist
both delay-independent criteria and delay-dependent criteria, of
which the latter one is proved to be less conservative and therefore
more important in practical applications, although sometimes more
challenging to obtain (see \cite{Moon}). The objective of this paper
is to achieve delay-dependent stability analysis of delay switched
stochastic systems with uncertainties. The delay-dependent stability
criteria are given in terms of linear matrix inequalities (LMIs) and
average dwell time.



\section{Problem statement and preliminaries}\label{sec2}
Consider the following stochastic uncertain switched system
\begin{align}
dx(t)&=[(A_i+\Delta A_i)x(t)+(\tilde{A_i}+\Delta\tilde{
A_i})x(t-h)]dt\nonumber\\
&\quad+[(B_i+\Delta B_i)x(t)+(\tilde{B_i}+\Delta\tilde{
B_i})x(t-h)]dw(t),\notag\\
&\qquad\qquad\text{for}\quad
t\geq0,\,\alpha(t)=i,\label{sw_sy}
\\[1\jot]
x(t)&=\phi(t),\,t\in [-h,0],\nonumber
\end{align}
where $x\in\Real^n$ is the state and $h$ is the constant time-delay.
$w(t)$ is a scalar Brownian motion. $\phi$ is the initial data.
$A_i, \tilde{A_i}\in\Real^{n\times n}$, $B_i$ and
$\tilde{B_i}\in\Real^{n\times m}$ are constant real matrices. Let
the switching signal $\alpha:\Real^+\rightarrow
I_K\overset{\triangle}{=}\{1,2,...,K\}$ be a piecewise constant and
continuous from the right deterministic function, with finitely many
discontinuities on every bounded subinterval of $\Real^+$. Without
loss of generality, we can suppose that $\alpha(t)$ is right
continuous at its points of discontinuity. $\Delta A_i$,
$\Delta\tilde{A_i}$, $\Delta B_i$ and $\Delta\tilde{B_i}$ are
time-dependent uncertainties of the form
\begin{align*}
\Delta A_i^T\Delta A_i\leq \widehat{A_i},\quad
\Delta\tilde{A_i}^T\Delta\tilde{A_i}\leq\widehat{\tilde{A_i}},
\\
\Delta B_i^T\Delta B_i\leq\widehat{B_i},\quad
\Delta\tilde{B_i}^T\Delta\tilde{B_i}\leq\widehat{\tilde{B_i}},
\end{align*}
where $\widehat{A_i}$, $\widehat{\tilde{A_i}}$, $\widehat{B_i}$, and
$\widehat{\tilde{B_i}}$ are known constant real matrices of
appropriate dimensions, which describe the uncertainty bounds. We
shall call this kind of uncertainties \emph{admissible
uncertainties}. Note that the assumption on the switching signal
$\alpha$ guarantees that system (\ref{sw_sy}) switches only finite
times on any bounded time interval. It is well-known that, given any
initial data $\phi\in L_{\mathcal{F}_0}^{2,h}$, any single system in
(\ref{sw_sy}) has a unique solution (see \cite{Maobook}). Based on
the assumption on the switching signal, it follows step by step that
the switched system (\ref{sw_sy}) has a unique solution as well,
denoted by $x(t;\phi)$.

Two types of stability are considered in this paper, one is almost
sure exponential stability and the other is exponential stability in
$p$th moment (see \cite{Maobook} for definitions). The main problem
now can be formulated as follows:
\begin{prob}
For all admissible uncertainties, under what conditions will the
uncertain switched system $(\ref{sw_sy})$ be almost surely
exponentially stable? Under what conditions will it be exponentially
stable in mean square?
\end{prob}

\section{Main results}\label{sec3}
In this section, the stability analysis of system (\ref{sw_sy}) is
studied.
\begin{thm}\label{main_thm_1}\
Given $h>0$, the uncertain switched stochastic system
$(\ref{sw_sy})$ is robustly exponentially stable in mean square if
the following two conditions hold:
\begin{enumerate}
\item  For each $i\in I_K$, there exist positive numbers
$\varepsilon_{1,i}$, $\varepsilon_{2,i}$, \ldots,
$\varepsilon_{9,i}$ and $\rho_i$, positive definite matrices $Q_i$,
$H_i$, $P_i$, and arbitrary matrices $G_i$ such that the following
matrix inequalities are satisfied:
\begin{align}
\Psi_i=\begin{bmatrix}
\Psi_{11}&\Psi_{12}&\Psi_{13}\\
\star&\Psi_{22}&\Psi_{23}\\
\star&\star&\Psi_{33}
\end{bmatrix}&<0,\label{ineq11}\\
Q_i&\leq\rho_iI\label{ineq12},
\end{align}

\item Let $T_a$ be the average dwell time of system
(\ref{sw_sy}),
\begin{equation}\label{switch_length_1}
T_a>\frac{\ln\beta}{\eps},
\end{equation}
where

and $\eps$ is the unique root of the equation
\begin{equation}\label{eqn}
\lambda-\eps c_2-c_3\eps h e^{\eps h}=0.
\end{equation}
\end{enumerate}
\end{thm}


The following proposition, which can be proved directly by Schur's
lemma \cite{Boyd}, gives an equivalent version of inequality
$(\ref{ineq11})$ which can be readily computed with Matlab's LMI
control toolbox \cite{Gahinet}.
\begin{prop}\label{ready_comp_ineq_1}The matrix
inequality $(\ref{ineq11})$ is equivalent to the following
inequality:
\begin{align}
\Lambda_i=
\begin{bmatrix}
\Lambda_{11} & \Lambda_{12} & \Lambda_{13}  & 0           & 0           & \Lambda_{16} \\
\star        & \Lambda_{22} & \Lambda_{23}  & \Lambda_{24}& 0           & 0          \\
\star        & \star        & \Lambda_{33}  & 0           & \Lambda_{35}& 0        \\
\star        & \star        & \star         & \Lambda_{44}& 0           & 0        \\
\star        & \star        & \star         & \star       & \Lambda_{55}& 0        \\
\star        & \star        & \star         & \star       & \star       & \Lambda_{66} \\
\end{bmatrix}&<0,
\end{align}
where
\begin{align*}
\Lambda_{11}&=H_i+hP_i+Q_i(A_i+G_i)+(A_i^T+G_i^T)Q_i^{}\\
&\quad+(\eps_{1,i}+h\eps_{3,i})\widehat{A_i}+B_i^TQ_iB_i\\
&\quad+(\rho_i+\eps_{5,i}+\eps_{7,i}+\eps_{8,i}^{-1}\rho_i^2)\widehat{B_i},\\
\Lambda_{12}&=Q_i(\tilde{A_i}-G_i)+B_i^TQ_i\tilde{B_i},\\
\Lambda_{22}&=-H_i+(\eps_{2,i}+h\eps_{4,i})\widehat{\tilde{A_i}}+\tilde{B_i}^TQ_i\tilde{B_i}\\
&\quad+(\rho_i+\eps_{6,i}+\eps_{8,i}+\eps_{9,i})\widehat{\tilde{B_i}},\\
\Lambda_{13}&=h(A_i^T+G_i^T)Q_iG_i,\\
\Lambda_{23}&=h(\tilde{A}_i^T-G_i^T)Q_iG_i,\\
\Lambda_{33}&=-hP_i,\\
\Lambda_{24}&=\begin{bmatrix}\tilde{B_i}^TQ_i &
\tilde{B_i}^TQ_i\end{bmatrix},\\
\Lambda_{44}&=\diag{-\eps_{7,i}I,-\eps_{9,i}I},\\
\Lambda_{35}&=\begin{bmatrix}hG_i^TQ_i^{} &
hG_i^TQ_i^{}\end{bmatrix},\\
\Lambda_{55}&=\diag{-\eps_{3,i}hI,-\eps_{4,i}hI},\\
\Lambda_{16}&=\begin{bmatrix}Q_i & Q_i & B_i^TQ_i &
B_i^TQ_i\end{bmatrix},\\
\Lambda_{66}&=\diag{-\eps_{1,i}I,-\eps_{2,i}I,-\eps_{5,i}I,-\eps_{6,i}I}.
\end{align*}
\end{prop}

In the previously stated results, we used multiple Lyapunov
functionals to deal with switched systems. If a single $Q$ can be
found to satisfy inequalities $(\ref{ineq11})$ and $(\ref{ineq12})$
, we have the following corollary of Theorem \ref{main_thm_1}.
\begin{cor}\label{single_P_1}
Given $h>0$, if there exists a positive definite matrix $Q$ such
that the matrix inequalities $(\ref{ineq11})$ and $(\ref{ineq12})$
are satisfied with $Q_i=Q$, then, for any given switching law, the
uncertain switched stochastic system $(\ref{sw_sy})$ is robustly
exponentially stable in mean square.
\end{cor}

\begin{rem} For the almost sure exponential stability, according to \cite{Mao} or
\cite{Maobook}, it can be shown that if the conditions of Theorem
\ref{main_thm_1} are satisfied, then system (\ref{sw_sy}) is almost
surely robust exponentially stable as well. Since under the given
conditions almost sure exponential stability can be easily derived
from exponential stability in mean square, we shall simply state a
system is exponential stability in the following section.
\end{rem}

\begin{rem}To conclude this section, we point out that, for briefness of
this paper, we only focus on system with linear uncertainties.
However, stability results for system with nonlinear uncertainties
can be proved via a similar approach.
\end{rem}

\section{Application}\label{sec4}
In this section, we present an example to illustrate our main
results. The following example shows that sometimes only multiple
Lyapunov functionals would work for the stability analysis.
\begin{ex}\label{ex2}Consider the switched system given by
\begin{align}
dx(t)&=\left[(A_1+\Delta A_1)x(t)+(\tilde{A}_1+\Delta\tilde{A}_1)x(t-h)\right]dt\notag\\
&\quad+g_1(t,x(t),x(t-h))dw(t),\quad\,\alpha(t)=1,\label{sw3}\\
\text{and}\nonumber\\
dx(t)&=\left[(A_2+\Delta
A_2)x(t)+(\tilde{A}_2+\Delta\tilde{A}_2)x(t-h)\right]dt\notag\\
&\quad+g_2(t,x(t),x(t-h))dw(t),\quad\,\alpha(t)=2\label{sw4},
\end{align}
where
\begin{align*}
A_1&=\begin{bmatrix}-1.9 & -3.3\\
7.6 & -11.3
\end{bmatrix},\quad\tilde{A_1}=\begin{bmatrix}-4 & 8.9\\
1.8 & -2.5
\end{bmatrix},\\
A_2&=\begin{bmatrix}-5.4 & 3.9\\
0 & -7.2
\end{bmatrix},\quad\tilde{A_2}=\begin{bmatrix} 5 & 2.6\\
-2.2 & -2.1
\end{bmatrix},
\end{align*}
\begin{align*}
\Delta A_1^T\Delta A_1&\leq\widehat{A}_1=0.01I,\quad
\Delta\tilde{A}_1^T\Delta\tilde{A}_1\leq\widehat{\tilde{A}}_1=0.01I,\\
\Delta A_2^T\Delta A_2&\leq\widehat{A}_2=0.01I,\quad
\Delta\tilde{A}_2^T\Delta\tilde{A}_2\leq\widehat{\tilde{A}}_2=0.01I,
\end{align*}
and
\begin{align*}
&\tr{g_1^T(t,x(t),x(t-h))g_1(t,x(t),x(t-h))}\\
&\quad\leq0.1\norm{x(t)}^2+0.1\norm{x(t-h)}^2,\\
&\tr{g_2^T(t,x(t),x(t-h))g_2(t,x(t),x(t-h))}\\
&\quad\leq0.1\norm{x(t)}^2+0.1\norm{x(t-h)}^2.
\end{align*}

\begin{table}
\caption{Stability bounds of time-delay and average dwell time
(Example \ref{ex2})}\label{table2}
\begin{center}
{\tt
\begin{tabular}{|c||c|c|c|c|}
\hline $T_0\,\,(\times 10^3)$& $0$ & $0.0856$ & $0.2303$ & $0.7960$\\
\hline $h_0$& $0$ & $0.5$ & $1.0$ & $1.5$\\
\hline $T_0\,\,(\times 10^3)$ &$1.7645$& $3.0717$ & $5.1803$ & $9.2734$\\
\hline $h_0$& $2.0$&$2.5$ & $3.0$ & $3.5$\\
\hline $T_0\,\,(\times 10^3)$ &$23.4052$& $238.2837$ & $$ & $$\\
\hline $h_0$& $4.0$&$4.4$ & $$ & $$\\
\hline
\end{tabular}
}
\end{center}
\end{table}

It can be shown using Matlab's LMI control toolbox that there exists
no single $Q$ to suit the inequalities (\ref{ineq11}) and
(\ref{ineq12}). However, if we apply Theorem \ref{main_thm_1}
instead of Corollary \ref{single_P_1}, we can see that the switched
system given by systems (\ref{sw3}) and (\ref{sw4}) is robustly
exponentially stable for $0\leq h\leq 4.4$. For different average
dwell time lower bounds $T_0$, the delay upper bounds $h_0$
guaranteeing the exponential stability of the system are listed in
Table \ref{table2}.
\end{ex}


\section*{Acknowledgements}
The research for this work was supported, in part, by the Natural
Sciences and Engineering Research Council of Canada.

\footnotesize

% Bibliography----------------------------------------------------------------



\end{document}
% ----------------------------------------------------------------
